\begin{table}%t3
\caption{\label{refveltab}References for independent velocity determinations.}
%\centering
\small
\begin{tabular}{ll}
\hline
\hline
BD99:	& Banks et~al. (\cite{Bank99}) \\
CF97:	& Cot\'e et~al. (\cite{Cote97}) \\
DN96:	& Di Nella et~al. (\cite{Dinel96}) \\
DN97:	& Di Nella et~al. (\cite{Dinel97}) \\
DT90:	& Djorgovski et~al. (\cite{Djor90}) \\
D91:    & Dressler (\cite{Dres91}) \\
F83:	& Fairall (\cite{Fair83}) \\
FW98:	& Fairall et~al. (\cite{Fair98}) \\
FW06:   & Fairall \& Woudt (\cite{Fair06}) \\
FH95:   & Fisher et~al. (\cite{Fish95}) \\
HW00:	& Hasegawa et~al. (\cite{Hase00}) \\
HIZSS:  & Henning et~al. (\cite{Henn00}) \\
HIPASS: & HIPASS (2006) \\
HG95:   & Huchra et~al. (\cite{Huch95}) \\
HK01:   & Huchtmeier et~al. (\cite{Huch01}) \\
JS00:   & Juraszek et~al. (\cite{Jura00}) \\
KD04:   & Koribalski \& Dickey (\cite{Kori04}) \\
HIPASS-BGC & Koribalski et~al. (\cite{Korib04}) \\
MF96:	& Mathewson \& Ford (\cite{Math96}) \\
MH91:   & Mould et~al. (\cite{Moul91}) \\
PT03:   & Paturel et~al. (\cite{Patu03}) \\
RS06:   & Radburn-Smith et~al. (\cite{Radb06}) \\
RK02:   & Ryan-Weber et~al. (\cite{Ryan02}) \\
SE95:   & Sanders et~al. (\cite{Sand95}) \\
SH92:	& Strauss et~al. (\cite{Stra92}) \\
TH07:   & Theureau et~al. (\cite{Theu07}) \\
RC3:    & de Vaucouleurs et~al. (\cite{devau91}) \\
Vv92:   & Visvanathan \& van den Bergh (\cite{Visv92}) \\
VY96:	& Visvanathan \& Yamada (\cite{Visv96}) \\
WK99:	& Woudt et~al. (\cite{Woud99}) \\
WK04:	& Woudt et~al. (\cite{Woud04}) \\
\hline
\end{tabular}
\end{table}